\begin{thebibliography}{10}

\bibitem{adar00free}
Eytan Adar and Bernardo~A. Huberman.
\newblock Free riding on gnutella.
\newblock {\em First Monday}, September 2000.

\bibitem{BAJ99}
Albert-La\'azl\'o, R\'eka Albert, and Hawoong Jeong.
\newblock Scale-free characteristics of random netwoks: The topology of the
  world wide web.
\newblock {\em submitted to {E}lsevier {Preprint}}, August 1999.

\bibitem{beth94valuation}
T.~Beth, M.~Borcherding, and B.~Klein.
\newblock Valuation of trust in open networks.
\newblock {\em Lecture Notes in Computer Science}, 875:3--??, 1994.

\bibitem{Bla96}
Matt Blaze.
\newblock Oblivious key escrow.
\newblock In {\em Proc. Workshop on Information Hiding}, number 1174 in LNCS,
  pages 334--343. Springer-Verlag, 1996.

\bibitem{BFL96}
Matt Blaze, Joan Feigenbaum, and Jack Lacy.
\newblock Decentralized trust management.
\newblock In {\em Proc. 17th Symposium on Security and Privacy}, pages
  164--173, Los Alamitos, 1996. IEEE Computer Society Press.

\bibitem{Bra97}
Marc Branchaud.
\newblock A survey of public key infrastructures.
\newblock Master's thesis, McGill University, Dept. of Computer Science, March
  1997.

\bibitem{CL99}
Miguel Castro and Barbara Liskov.
\newblock Practical byzantine fault tolerance.
\newblock In {\em Proc. Third Symp. on Operating Systems Design and
  Implementation}, New Orleans, February 1999.

\bibitem{chaum89untraceable}
D.~Chaum, A.~Fiat, and M.~Naor.
\newblock Untraceable electronic cash (extended abstract), 1989.

\bibitem{crary99typed}
Karl Crary, David Walker, and Greg Morrisett.
\newblock Typed memory management in a calculus of capabilities.
\newblock In {\em Conference Record of {POPL} 99: The 26th {ACM}
  {SIGPLAN}-{SIGACT} Symposium on Principles of Programming Languages, San
  Antonio, Texas}, pages 262--275, New York, NY, 1999.

\bibitem{chord:hotos}
Frank Dabek, Emma Brunskill, M.~Frans Kaashoek, David Karger, Robert Morris,
  Ion Stoica, and Hari Balakrishnan.
\newblock Building peer-to-peer systems with {C}hord, a distributed lookup
  service.
\newblock In {\em Proceedings of the 8th {W}orkshop on {H}ot {T}opics in
  {O}perating {S}ystems ({HotOS-VIII})}, Schloss Elmau, Germany, May 2001.
  {IEEE} {C}omputer {S}ociety.

\bibitem{dingledine00free}
Roger Dingledine, Michael~J. Freedman, and David Molnar.
\newblock The free haven project: Distributed anonymous storage service.
\newblock In {\em Workshop on Design Issues in Anonymity and Unobservability},
  number 2009 in LNCS, pages 67--95, 2000.

\bibitem{dingledine-cascrep}
Roger Dingledine and Paul Syverson.
\newblock Reliable {MIX} cascade networks through reputation.
\newblock In {\em Proc. Financial Cryptography}, March 2002.

\bibitem{douceur-sybil}
John~R. Douceur.
\newblock The sybil attack.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{dwork-spam}
Cynthia Dwork and Moni Naor.
\newblock Pricing via processing or combating junk mail.
\newblock In Ernest~F. Brickell, editor, {\em Advances in Cryptology -- CRYPTO
  '92}, number 740 in LNCS, pages 139--147. Springer-Verlag, 1992.

\bibitem{flake02self-organization}
Gary~William Flake, Steve Lawrence, C.~Lee Giles, and Frans Coetzee.
\newblock Self-organization of the web and identification of communities.
\newblock {\em IEEE Computer}, 35(3):66--71, 2002.

\bibitem{gay01language}
David Gay and Alexander Aiken.
\newblock Language support for regions.
\newblock In {\em {SIGPLAN} Conference on Programming Language Design and
  Implementation}, pages 70--80, 2001.

\bibitem{kleinberg-smallworld}
Jon Kleinberg.
\newblock The small-world phenomenon: An algorithmic perspective.
\newblock In {\em Proc. 32nd ACM Symp. on Theory of Computing}, 2000.

\bibitem{kleinberg99clever}
Jon~M. Kleinberg.
\newblock Authoritative sources in a hyperlinked environment.
\newblock {\em Journal of the {ACM}}, 46(5):604--632, 1999.

\bibitem{LSP82}
Leslie Lamport, Robert Shostak, and Marshall Pease.
\newblock The byzantine generals problem.
\newblock {\em {ACM} Transactions on Programming Languages and Systems},
  4(3):382--401, July 1982.

\bibitem{LA98}
Raph Levien and Alexander Aiken.
\newblock Attack resistant trust metrics for public key certification.
\newblock In {\em 7th {USENIX} Security Symposium}, San Antonio, Texas, January
  1998.

\bibitem{Mau96}
Ueli Maurer.
\newblock Modelling a public-key infrastructure.
\newblock In E.~Bertino, H.~Kurth, G.~Martella, and E.~Montolivo, editors, {\em
  Computer Security -- {ESORICS} '96}, number 1146 in LNCS. Springer Verlag,
  1996.

\bibitem{maymounkov-kademlia}
Petar Maymounkov and David Mazi\`eres.
\newblock Kademlia: A peer-to-peer information system based on the {XOR}
  metric.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{agorics-incentive}
Mark~S. Miller and K.~Eric Drexler.
\newblock Incentive engineering: for computational resource management.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  231--266. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{agorics-markets}
Mark~S. Miller and K.~Eric Drexler.
\newblock Markets and computation: Agoric open systems.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  133--176. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{moreton-trust}
Tim Moreton and Andrew Twigg.
\newblock Trading in trust, tokens and stamps.
\newblock In {\em Proc. Workshop on Economics of Peer-to-Peer Systems}, June
  2003.

\bibitem{P2P}
Andy Oram, editor.
\newblock {\em Peer to Peer}.
\newblock O'Reilly \& Associates, 2001.

\bibitem{page98pagerank}
Lawrence Page, Sergey Brin, Rajeev Motwani, and Terry Winograd.
\newblock The {P}age{R}ank citation ranking: Bringing order to the web.
\newblock Technical report, Stanford University, 1998.

\bibitem{Pos99}
David Post, 1999.

\bibitem{operatingthetan}
ptsc (pseudonym).
\newblock The {C}hurch of {S}cientology's supremacy over the search term
  ``{S}cientology'' on {G}oogle, February 2002.

\bibitem{RS97a}
Michael Reiter and Stuart Stubblebine.
\newblock Path independence for authentication in large-scale systems.
\newblock In {\em Proceedings of the 4th {ACM} Conference on Computer and
  Communications Security}, 1997.

\bibitem{RS97b}
Michael Reiter and Stuart Stubblebine.
\newblock Toward acceptable metrics of authentication.
\newblock In {\em Proceedings of the 1997 {IEEE} Symposium on Security and
  Privacy}, 1997.

\bibitem{Ross67}
D.~T. Ross.
\newblock The {AED} free storage package.
\newblock {\em Communications of the ACM}, 10(8):481--492, August 1967.

\bibitem{chord:sigcomm01}
Ion Stoica, Robert Morris, David Karger, M.~Frans Kaashoek, and Hari
  Balakrishnan.
\newblock Chord: A scalable peer-to-peer lookup service for internet
  applications.
\newblock In {\em Proceedings of the {ACM} {SIGCOMM} '01 Conference}, San
  Diego, California, August 2001.

\bibitem{tarah92associating}
Anas Tarah and Christian Huitema.
\newblock Associating metrics to certification paths.
\newblock In {\em European Symposium on Research in Computer Security
  ({ESORICS})}, pages 175--192, 1992.

\bibitem{tofte97regionbased}
Mads Tofte and Jean-Pierre Talpin.
\newblock Region-based memory management.
\newblock {\em Information and Computation}, 1997.

\bibitem{zooko-mn}
Bryce Wilcox-O'Hearn.
\newblock Experiences deploying a large-scale emergent network.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\end{thebibliography}

\chapter{Introduction}
\label{intro}

In today's world of open, decentralized networks, the question of
\emph{trust} is becoming increasingly relevant. Most existing Internet
protocols implement a naive policy of providing a relatively limited
set of services, but trusting all users with them. Given that a
significant fraction of Internet users are not trustworthy, the
inevitable result is spam, denial of service attacks, cracking, and a
host of other ills far too familiar to legitimate Internet users.

A number of approaches have been developed and deployed over the
years, with varying degrees of success. The most basic is the model of
password-protected accounts. This mechanism is almost universally
deployed, but suffers from serious limitations, including the need for
servers to manage the accounts, the need for users to keep track of a
large number of passwords, and the relative lack of security provided
by this model. Thus, there has been a sustained interest in more
sophisticated models.

One such approach is the \emph{Public Key Infrastructure}, or PKI
\cite{Bra97}. Briefly, a PKI consists of various \emph{Certification
Authorities} (or CA's) that issue \emph{certificates} asserting a
binding between a name and a public key. PKI's suffer from two
fundamental problems: the lack of useful meaning in the PKI's
underlying namespace, and the question of which CA to trust. Actual
implementations of CA's have proved themselves not worthy of absolute
trust. Further, as the number of CA's deployed scales up, the risk of
any one of them being compromised scales accordingly. In part because
of these two problems, PKI's have met with limited success at best.

Spurred on by these limitations, much recent attention has been
focussed on the explicit encoding and evaluation of trust
relationships. A pioneering work in this area is the PolicyMaker
framework of Matt Blaze\cite{BFL96}. A particularly interesting branch
of this work is the concept of \emph{trust metrics,} which are the
primary focus of this thesis.

Trust metrics are based on the principle of \emph{local} encoding of
trust relationships, but granting trust on a global scale. These
assumptions closely parallel the philosophy of peer to peer
networks\cite{P2P}.

While the literature is rich with design proposals for trust metrics,
there has been relatively little analysis of how well they work in
practice. In Chapter~\ref{tmetrics}, we show that, under assumptions
similar to the Internet, the entire category of ``scalar'' trust
metrics fails to resist easily-mounted attacks. Attacks against these
poor trust metrics dot the literature\cite{RS97b}, and have been used
to argue (incorrectly) that a centralized identity service is
needed\cite{douceur-sybil}.

While the outlook for scalar trust metrics is indeed bleak, we propose
a new class, which we call \emph{group} trust metrics, so called both
because the trust metric is very well suited to evaluating membership
in a group, and because this evaluation is done over the entire group
of nodes, rather than individually for each node. While the group
trust metric cannot prevent individual hostile nodes from being
accepted as group members, it can place strict bounds on the
\emph{number} of hostile nodes so accepted.

I built a community website for free software developers, called
Advogato, that uses this group trust metric to determine who belongs
to the community. Acceptance by the trust metric confers privileges to
post articles and comments, and to edit project information. At the
time of this writing, the site has been in operation for about 18
months, and has built a trust graph of over 1000 active nodes.
Advogato is notable for the extremely low level of trolls, spam, and
other forms of abuse common to bulletin board type systems, thus
providing strong anecdotal evidence that the trust metric is
effective. Advogato is described in more detail in
Chapter~\ref{advogato}.

One interesting application of trust metrics is secure registration
and lookup of public keys bound to names, essentially the same problem
addressed by Public Key Infrastructures (PKI). The attack resistant
properties of the trust metric avoid the single point of vulnerability
common to most PKI designs. We present a detailed design for such an
attack-resistant name service in Chapter~\ref{nameservice}. The desire
to build a ``better PKI'' was the original motivation behind the work
of this thesis.

Because the group trust metric is effective in resisting attack, it
has many other interesting applications. One such is an
attack-resistant system for distributing metadata, for example
opinions about songs. Attack resistance can be useful for ensuring
high quality of the metadata, but becomes particularly important when
there are financial implications to the metadata system, for example
to identify recipients of voluntary donations to artists who post
their music to the Net. We propose a design for an attack resistant
metadata system in Chapter~\ref{metadata}. This design combines ideas
from both the Advogato group trust metric and the PageRank algorithm
used in Google.

Another intriguing application for group trust metrics is to provide a
spam resistant infrastructure for e-mail. Spam is a huge problem,
causing lots of wasted time for many, and most e-mail users feel
hopeless in the face of it. We present a partial design for a
communications infrastructure resistant to spam, yet highly permeable
to legitimate email. This design uses a capacity constrained flow
network where ``stamps'' are the commodity of flow. Rather than
explicitly computing a flow in the network, the flow results from
local activity. When capacity is exhausted, e-mail is no longer
deliverable. This design has many appealing properties, but several
aspects remain as open problems, particularly scaling.
Chapter~\ref{stamptrading} presents this design and discusses some
open issues.
